EFOG TURNOVER                 Date _6/1/2021__    
Present: Nena, Liz, Steve, Bruno, Shane, Bruce, Katy, Matt, John, Wendy, Ashley, Clay
[bookmark: _Hlk39760986][bookmark: _Hlk39761201]
EFOG Presence onsite: 5/31/21-6/6/21
Mon-Fri FC1: Ashley  FC2: John
M,Tues Evenings – Ashley Wed – Fri evenings: Bruno
Weekend:  Ashley-OC
Vacation/DORs:  Nena 6/2/21
Monday 5/31 is a lab holiday
      
Things to look out for this week:
· 5-ID-A has a portable O2 monitor on the hutch door.  Long vendor delays for a new O2 monitor are requiring us to continue this periodic portable O2 monitor check.  Please check for normal O2 levels (20.9%) and battery level ~ 3 times per week for on-shift personnel and please report to me if there is a problem.
· Issued GasBadge portable oxygen monitor #14 at the 20-ID-A station door since the wall mounted unit needs repairs/replacement. All relevant warning signs are posted to warn of its use. Users need to use the portable O2 monitor every time upon entering the 20-ID-A station. The portable unit will be in use until the permanent wall mounted unit (series 1000) is replaced/repaired.
· Eyewashes/LSI. Eyewashes are due weekly and  LSI are due at the end of the month. 

Events from last week
· Ongoing shutdown activities
Ongoing work on experimental floor – RSS
· See Shutdown CCWP sheet
Ongoing work on experimental floor/LOMS – Other
·  

Completed Repairs and Work
· LOM 438 Parking lot repaving has been completed.

Upcoming Work on experimental floor/LOMs
· Sector 4, 6 & others: -Helium Recovery System-See notes from 3/29 
· Sector 5 –Gas cabinet installation-See notes from 3/8 (for dark period)

Pre-RUN Start up
· FC Pre-Run Checklist
· 2021 April_May Shutdown folder
· 2021-2 Shutdown CCWP and Start-up
· 2021-2 Commissioning ESAFs
· Pre-Run Briefing Tentatively planned for Tuesday, June 1

28-ID IDEA Status 
· Equipment installation on-going.

25-ID ASL Status
· Caratelli finishing by May 18-utilities possibly in June
· Tuesday, 5/18, at 9 a.m. at 25-ID for the punch list walk-through

LBB Updates
· Activities This Week: Deck edge HEXM and ISN, safety cables, decking, walls.  CMU wall installations and ISN concrete work. Mechanical mezzanine patching and painting. More equipment to be lifted, but not yet on-site.  
· Equipment lifts inside the mezzanine will wrap on 6/1 and will be scheduled on Mondays/machine studies during the 2021-2 run.
· Items to Keep an Eye on:  none
· Looking Ahead: NA
· Scheduling:  Typically contractors work from 7:00 a.m. - 3:30 p.m.   LBB updates can be found at https://www.aps.anl.gov/Machine-Status/Construction-Schedule/long-beamline-building-construction-schedule

437 lab build out
· Funding issue has been resolved! Contractors set to start in August.
· Clay will work with ANL and contractor and be the POC for our group

OPS/XSD GL/PSC Managers/Leader Meeting: 
· .

Experiments of Note 

· Limited Operations + with approved ANL onsite users from divisions outside of PSC and some onsite external users with PSC ALD and DOE ASO approval.
· All risk level work allowed except for Radioactive, Explosive, and BSL2 samples unless the experiment is deemed mission critical or funding dependent, etc.  Most experiments are mail-in or remote.
· No plans for change in operation status until after 2021-2 run.

ESAFs with onsite users from ANL divisions outside of PSC have been approved by DOE ASO.  ESAFs with external (outside of ANL) onsite users have been approved by both PSC ALD and DOE ASO.

Experiments Running 6/1/2021 - 6/7/2021

Special Experiments:

· ESAF 241628 at 7-BM-B.  Travis Sikes and Robert Tranter from CSE onsite at APS.  Start 6/2/2021.  End 6/6/2021.  ESAF is conditionally approved as an open flame permit must be issued by the ANL FD.  ESAF will be changed to full approval once the open flame permit has been issued.
· ESAF 241845 at 20-ID-C.  Experiment is using a sealed source in the beam path and thus is considered a radioactive sample.  Steve Heald and Zou Frinfrock onsite.  Start 6/3/2021.  End 6/5/2021.
· ESAF 241265 at 16-ID-B.  Zsolt Jenei and Jae Klepeis from LLNL onsite at APS.  Start 6/3/2021.  End 6/5/2021.  Rad Samples

Long Duration Lab or No X-Ray ESAFs:

· ESAF 239935 at Sector 7.  Chris Otolski and Gilles Doumy from CSE onsite at APS.  Start 5/4/2021.  End 8/3/2021.
· ESAFs 241561 and 240825 at HP-CAT lab.  Ravhi Kumar and Nilesh Salke from UIC onsite at APS.  Start 5/28/2021.  End 6/24/2021.

Long Term Work Status - APS Upgrade and other APS projects:  See EFOG Turnover notes from previous week for more information.

Round Table Discussion

Nena  
· See EFOG Turnover notes (below) for further details on Ops, LBB, Shutdown work, general info, etc.
· Saliva COVID Testing:  See 3/29 turnover notes for dates and times.
· IF you are fully vaccinated (2 weeks after last shot) you may opt out of weekly saliva testing- please see emailed memo for more details and process
· Mask mandate for fully vaccinated individuals (ANL employees, Argonne Associates, Contractors like Scheck, and CAT beamline personnel) has been lifted.  A copy of a person’s vaccination card must be uploaded to Box Plus.
· Users from outside of ANL must still wear masks regardless of vaccination status
· Cannot ask others their vaccination status.
· Occupancy numbers in hutches, etc have not changes
· Increasing EFOG presence on site: if you are not on shift for the week please be on site at least 3 days a week. 
· Hybrid work model is being discussed.  GLs will make group recommendations by 5/28.  Individual discussions will occur around June 15
· Updated EFOG Shift Schedule to include April/May Shutdown and 2021-2 RUN
· FC 2021-2 Schedule
· DOR Calendar
· A new folder has been created in EFOG Shared in Box entitled Floor Coordinator Responsibilities and Information. 
· FC ON Shift Responsibilities
· FC Shift Exchange Template for MCR
· Floor Coordinator Responsibilities for Three-Year Sector Safety Reviews
· Troubling Shooting FC1/FC2 Paging System
· Construction on West Gate to begin April 15- Aug 15
· Will be closed to all traffic from 9am to 3pm during this time
· Met with FAC last week to have our names put on WEC and WPCs; especially for those on the experiment floor
· Nena will work with Tim Smith to put in a key request for 436 E095
· Shane to work on setting up a computer for Tim

Matt and Katy will do Global Onlines (mid-afternoon on 6/1).
Bruno will enable all stations except for the A stations and then wait for word from Matt and Katy.

Katy will cover 9am-10am for Ashley as FC1 on 6/2.

Check to ensure BIVs and FIVs are open (if BLEPS vac permit isssued)

Katy
· See email from this morning for beamline status
· 1-BM will post their commissioning ESAF later this morning
· All stations can be APS enables
· Will check with 3-ID on time for Type C survey
Bruno
· No CCWPs
· All stations can be APS enabled
Clay
· CCCLs and QEWs signed
· Several Type Cs including MH doors at 9-BM, 11-ID, and 12-ID
John
· All beamline ready for APS enable except 16-BM
· 15-ID needs Type C on MH
Matt
· All beamlines ready for APS enable
· ESAFs posted, QEWs and CCCLs completed
Ashley
· All beamlines ready for APS enable except for 23-BM and 24-BM as they are not running
· Type C for MH needed at 23-ID
Shane
· 26 needs to post their commissioning ESAF
· Type C at 28-ID-B
· Caratelli has left the APS
Steve
· Need to post several commissioning ESAFs
· Type C at 30-ID
Wendy
· Type Cs on Wednesday; 3 are MH
· Whoever does Global Onlines – Put the 28-ID key in the lock but do not turn the key
Liz
· Refer to spreadsheet in box for commissioning ESAF posting status
· See experiments of note section; ESAF at 7-BM with open flame permit and 2 rads this week
· PSC ALD and DOE ASO approvals still required where applicable
	
	
	



